# Duration of intravenous antibiotic therapy for children with acute osteomyelitis or septic arthritis: a feasibility study

Hans de Graaf,<sup>1,2</sup> Priya Sukhtankar,<sup>1,2</sup> Barbara Arch,<sup>3</sup> Nusreen Ahmad,<sup>1,2,4</sup> Amanda Lees,<sup>5</sup> Abigail Bennett,<sup>3</sup> Catherine Spowart,<sup>3</sup> Helen Hickey,<sup>3</sup> Annmarie Jeanes,<sup>6</sup> Kate Armon,<sup>7</sup> Andrew Riordan,<sup>8</sup> Jethro Herberg,<sup>9</sup> Scott Hackett,<sup>10</sup> Carrol Gamble,<sup>3</sup> Delane Shingadia,<sup>11</sup> Ann Pallett,<sup>2</sup> Stuart C Clarke,<sup>12</sup> Philip Henman,<sup>13</sup> Marieke Emonts,<sup>13</sup> Mike Sharland,<sup>14</sup> Adam Finn,<sup>15,16</sup> Andrew J Pollard,<sup>17</sup> Colin Powell,<sup>18,19</sup> Peter Marsh,<sup>20</sup> Claire Ballinger,<sup>21</sup> Paula R Williamson,<sup>3</sup> Nicholas MP Clarke<sup>2,22</sup> and Saul N Faust<sup>1,2\*</sup>

- <sup>1</sup>National Institute for Health Research Wellcome Trust Clinical Research Facility, University of Southampton, Southampton, UK
- <sup>2</sup>University Hospital Southampton NHS Foundation Trust, Southampton, UK <sup>3</sup>Medicines for Children Clinical Trials Unit, Clinical Trials Research Centre,
- Department of Biostatistics, University of Liverpool, Liverpool, UK <sup>4</sup>Public Health England, Southampton, UK
- <sup>5</sup>Health and Wellbeing Research and Development Group, University of Winchester, Winchester, UK
- <sup>6</sup>Radiology, Leeds Children's Hospital, Leeds Teaching Hospitals NHS Trust, Leeds, UK
- <sup>7</sup>Norfolk and Norwich University Hospitals NHS Foundation Trust, Norwich, UK <sup>8</sup>Alder Hey Children's NHS Foundation Trust, Liverpool, UK
- <sup>9</sup>Section of Paediatrics, Imperial College London, St Mary's Campus, London, UK <sup>10</sup>Heart of England NHS Foundation Trust, Birmingham, UK
- <sup>11</sup>Paediatric Infectious Diseases, Great Ormond Street Hospital for Children, London, UK
- <sup>12</sup>Academic Unit of Clinical and Experimental Sciences, Faculty of Medicine, University of Southampton, Southampton, UK
- <sup>13</sup>Newcastle upon Tyne Hospitals NHS Foundation Trust, Newcastle, UK
- <sup>14</sup>St George's University Hospitals NHS Foundation Trust, London, UK
- <sup>15</sup>Bristol Children's Vaccine Centre, School of Clinical Sciences, University of Bristol, Bristol, UK
- <sup>16</sup>Paediatric Infectious Diseases and Immunology, Bristol Royal Hospital for Children, Bristol, UK
- <sup>17</sup>Department of Paediatrics, Medical Sciences Division, University of Oxford, Oxford, UK

 <sup>18</sup>School of Medicine, University of Cardiff, Cardiff, UK
<sup>19</sup>Department of Paediatrics, University Hospital of Wales, Cardiff, UK
<sup>20</sup>Public Health England, South East Public Health England Regional Laboratory, Southampton, UK

<sup>21</sup>Faculty of Health Sciences, University of Southampton, Southampton, UK <sup>22</sup>Paediatric Orthopaedics, University of Southampton, Southampton, UK

\*Corresponding author s.faust@soton.ac.uk

Declared competing interests of authors: Saul N Faust was UK chief investigator and University Hospital Southampton NHS Foundation Trust principal investigator for a Cubist-sponsored clinical trial of daptomycin against standard of care antibiotic therapy in paediatric osteomyelitis. All funds were paid into accounts within the NHS trust or university and not paid as personal fees. He reports consultancy fees for advisory board participation paid into accounts within the NHS trust or university (not personal fees) from vaccine manufacturers and antimicrobial agent manufacturers, including AstraZeneca, Cubist, Merck, GlaxoSmithKline (GSK) and Pfizer, outside the submitted work. He acts as principal investigator for clinical trials and other studies conducted on behalf of University Hospital Southampton NHS Foundation Trust/ University of Southampton that are sponsored by vaccine manufacturers and antimicrobial agents, and has participated in advisory boards for vaccine manufacturers. Adam Finn reports grants and personal fees from Sanofi Pasteur MSD Ltd (SPMSD), and grants and personal fees from GSK, outside the submitted work. In addition, prior to October 2014, the University of Bristol and University Hospitals Bristol NHS Foundation Trust received funding for research conducted by Adam Finn and for consultancy and lectures from Pfizer, GSK, SPMSD and Novartis, who manufacture licensed and developmental meningococcal vaccines. Stuart C Clarke acts as principal investigator for clinical trials and other studies conducted on behalf of University Hospital Southampton NHS Foundation Trust/University of Southampton that are sponsored by vaccine manufacturers but receive no personal payments from them. He has participated in advisory boards for vaccine manufacturers and has received financial assistance from vaccine manufacturers to attend conferences, but receives no personal payments for this work. All grants and honoraria are paid into accounts within the respective NHS trusts or universities. Jethro Herberg reports that he was an investigator for a Cubist-sponsored clinical trial of daptomycin against standard of care antibiotic therapy in paediatric osteomyelitis; he received no personal payments. Andrew Riordan reports that the trust, but not he, has received funding for research sponsored by vaccine manufacturers and antimicrobial agents. He was lead editor for an e-learning package on meningitis produced by the Royal College of Paediatric and Child Health, funded by an unrestricted grant from Novartis Vaccines. Marieke Emonts was Newcastle upon Tyne Hospitals Foundation Trust's principal investigator for a Cubist-sponsored clinical trial of daptomycin against standard of care antibiotic therapy in paediatric osteomyelitis. All grants and honoraria are paid into accounts within the NHS trust or university; she received no personal payment of any kind. Marieke Emonts reports acting as site principle investigator on behalf of the Newcastle upon Tyne Hospitals Foundation Trust/Newcastle University for clinical trials sponsored by Merck Sharp & Dohme Corp. and SPMSD outside the submitted work. Claire Ballinger is currently a member of the Health Technology Assessment (HTA) Primary Care, Community and Preventive Interventions (PCCPI) Panel and the HTA PCCPI Methods group. She was a former member of the Research for Patient Benefit London committee. Claire Ballinger is supported by the National Institute for Health Research (NIHR) Collaboration for Applied Health Research and Care Wessex. Catherine Spowart reports grants from NIHR during the conduct of the study. Philip Henman reports working as an investigator in a commercial clinical trial sponsored by Merck Sharpe & Dohme, on behalf of his employer, outside the submitted work.

Published September 2017 DOI: 10.3310/hta21480

# **Plain English summary**

## IV antibiotic therapy in children with OM or SA

Health Technology Assessment 2017; Vol. 21: No. 48 DOI: 10.3310/hta21480

NIHR Journals Library www.journalslibrary.nihr.ac.uk

# **Plain English summary**

Children's bone and joint infections are treated first with intravenous antibiotics and then oral antibiotics. Historically, long courses of intravenous antibiotics were thought to be needed, which often involve a long hospital stay. At present there is no consensus regarding how long to treat these infections overall, or how long to use each of intravenous or oral antibiotics. This study used a combination of methods to assess if a clinical trial to reduce intravenous antibiotics in children will be feasible in the UK.

Data were collected from 44 UK centres to find out the current bacterial causes of children's bone and joint infections and how these are treated in clinical practice. A qualitative study explored families' views and experiences of having had a child treated for bone or joint infection. A consensus meeting was informed by information from the clinical data collection, web-based clinician survey and qualitative study to establish the final set of outcomes, oral switch criteria and overall future trial feasibility.

We found that clinicians are already switching from intravenous to oral antibiotics at an early stage in selected patients with simple disease, even though there is not yet formal clinical trial evidence of the safety of doing so. This has important implications both for current practice and for the feasibility of different potential trial designs. We also found that a future randomised controlled trial of shorter or longer courses of intravenous antibiotics would be feasible by including only children who need more than a minimum of 7 days' intravenous antibiotic treatment before switching to oral medicine.

## **Health Technology Assessment**

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 4.236

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, The Cochrane Library and the Clarivate Analytics Science Citation Index.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

#### Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

#### **HTA programme**

The HTA programme, part of the National Institute for Health Research (NIHR), was set up in 1993. It produces high-quality research information on the effectiveness, costs and broader impact of health technologies for those who use, manage and provide care in the NHS. 'Health technologies' are broadly defined as all interventions used to promote health, prevent and treat disease, and improve rehabilitation and long-term care.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

For more information about the HTA programme please visit the website: http://www.nets.nihr.ac.uk/programmes/hta

#### This report

The research reported in this issue of the journal was funded by the HTA programme as project number 10/146/01. The contractual start date was in March 2013. The draft report began editorial review in January 2016 and was accepted for publication in November 2016. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health.

© Queen's Printer and Controller of HMSO 2017. This work was produced by de Graaf *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

### Health Technology Assessment Editor-in-Chief

**Professor Hywel Williams** Director, HTA Programme, UK and Foundation Professor and Co-Director of the Centre of Evidence-Based Dermatology, University of Nottingham, UK

### **NIHR Journals Library Editor-in-Chief**

Professor Tom Walley Director, NIHR Evaluation, Trials and Studies and Director of the EME Programme, UK

### **NIHR Journals Library Editors**

**Professor Ken Stein** Chair of HTA and EME Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals)

Dr Martin Ashton-Key Consultant in Public Health Medicine/Consultant Advisor, NETSCC, UK

**Professor Matthias Beck** Chair in Public Sector Management and Subject Leader (Management Group), Queen's University Management School, Queen's University Belfast, UK

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Ms Tara Lamont Scientific Advisor, NETSCC, UK

**Dr Catriona McDaid** Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK

**Professor James Raftery** Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

**Professor Jim Thornton** Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

**Professor Martin Underwood** Director, Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of members of the NIHR Journals Library Board: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk